Author’s response to reviews

Title: Clinical and Biological Significance of STAT1 in Esophageal Squamous Cell Carcinoma

Authors:

Ying Zhang (ying18@ualberta.ca)
Ommoleila Molavi (molaviazar@yahoo.com)
Min Su (minsu@stu.edu.cn)
Raymond Lai (rlai@ualberta.ca)

Version: 4
Date: 7 October 2014

Author's response to reviews: see over
Clinical and Biological Significance of STAT1 in Esophageal Squamous Cell Carcinoma
Ying Zhang, Ommoleila Molavi, Min Su and Raymond Lai

Editor's comment
Requesting Ethics statement

Research involving human subjects (including human material or human data) that is reported in the manuscript must have been performed with the approval of an appropriate ethics committee. Research carried out on humans must be in compliance with the Helsinki Declaration (http://www.wma.net/en/30publications/10policies/b3/index.html). A statement to this effect must appear in the Methods section of the manuscript, including the name of the body which gave approval, with a reference number where appropriate.

Response: Modified as suggested. We have included a statement in the Method

Requesting consent statement:

Please state in the Methods section whether written informed consent for participation in the study was obtained from participants or, where participants are children, a parent or guardian.

Additional Request:
1. Please structure your abstract according to our guidelines: (http://www.biomedcentral.com/bmccancer/authors/instructions/researcharticle#formatting-abstract)
Response: Modified as suggested.
2. Please indicate the source of the cell lines used in your study in your methods.
Response: Modified as suggested.
3. Please include a conclusion section, according to our guidelines: (http://www.biomedcentral.com/bmccancer/authors/instructions/researcharticle#formatting-conclusions)
Response: Modified as suggested.

We sincerely hope that this manuscript is now acceptable for publication

Ying Zhang